A Survey on Intelligent Optimization Approaches to Boiler Combustion Optimization

Jing Liang^{1,2,3}, Hao Guo^{1,2}, Ke Chen^{1,2} [™], Kunjie Yu^{1,2}, Caitong Yue^{1,2}, and Yunpeng Ma⁴

ABSTRACT

This paper reviews the researches on boiler combustion optimization, which is an important direction in the field of energy saving and emission reduction. Many methods have been used to deal with boiler combustion optimization, among which evolutionary computing (EC) techniques have recently gained much attention. However, the existing researches are not sufficiently focused and have not been summarized systematically. This has led to slow progress of research on boiler combustion optimization and has obstacles in the application. This paper introduces a comprehensive survey of the works of intelligent optimization algorithms in boiler combustion optimization and summarizes the contributions of different optimization algorithms. Finally, this paper discusses new research challenges and outlines future research directions, which can guide boiler combustion optimization to improve energy efficiency and reduce pollutant emission concentrations.

KEYWORDS

boiler combustion optimization; circulating fluidized bed boiler; environmental protection; computational intelligence; intelligent optimization algorithm

ith the fast development of economy and technology in recent decades, China's electricity demand has increased. Figures 1 and 2 depict information from various forms of power generation in recent years. Statistics show that a sizable share of China's power output still comes from thermal power. The boiler carrier in the thermal power plant is the circulating fluidized bed boiler (CFBB). Coal has become the primary fuel for CFBB in most countries due to its low price and abundant reserves^[1, 2]. The pollution gas generated by coal combustion has seriously damaged the ecological environment. Many countries have formulated the corresponding laws and regulations to specify the pollution emission concentration of boilers^[3,4].

The use of coal-fired power generation mainly causes the following two results: (1) a lot of energy consumption and (2)



Fig. 1 Total power generation trend diagram of different power generation forms.



severe environmental pollution. Although coal reserves are relatively wealthy, they are not renewable resources. In addition, the current utilization rate of coal in thermal power plant boilers is relatively low, resulting in a waste of resources. Furthermore, a large number of harmful gases are generated after coal combustion, mainly including sulfide concentration (SO_x), carbon oxide (CO_x) , and nitrogen oxide concentration (NO_x) . These pollutants cause serious environmental problems such as acid rain, greenhouse effects, and the depletion of the earth's ozone layer. Therefore, how to effectively reduce pollutant emissions and enhance boiler thermal efficiency has essential significance for the sustainable development of the global economy. In the 1970s, research on boiler combustion optimization technology began. At first, Launder and Spalding^[5] combined mechanics and combustion knowledge to obtain the mathematical model and numerical calculation method of the combustion process, which

1 School of Electrical and Information Engineering, Zhengzhou University, Zhengzhou 450001, China

2 State Key Laboratory of Intelligent Agricultural Power Equipment, Luoyang 471000, China

3 School of Electrical Engineering and Automation, Henan Institute of Technology, Xinxiang 453000, China

© The author(s) 2023. The articles published in this open access journal are distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/).

⁴ School of Information Engineering, Tianjin University of Commerce, Tianjin 300134, China

Address correspondence to Ke Chen, chenkezixf@zzu.edu.cn

opened a new idea in combustion theory and applied research. Air cover is an effective method to ensure airflow uniformity in a CFBB boiler. In order to improve the thermal efficiency, Liu et al.^[6] designed a 220 t/h CFBB small test bench, integrated with a new gas cap. The mathematical model was deduced, and cold experiments verified the accuracy. The experimental results showed that the thermal efficiency of CFBB increased from 86.4% to 91.8% after using the new gas cap. The early research is mainly divided into three categories, namely physical and chemical methods, numerical simulation, and burner design optimization. Representative works are shown in Table 1.

Although these methods improve the thermal efficiency of the boiler and reduce pollutant emissions to a certain extent, with the continuous increase of boiler capacity, the combustion process becomes more complex, and it is difficult to use mechanical methods to optimize the boiler combustion process. Therefore, an efficient intelligent optimization technology is needed to better solve the boiler combustion optimization problem. Evolutionary computing (EC) technology and data modeling (DM) technology have attracted a lot of attention in boiler combustion optimization problems due to their efficient performance^[31, 32]. However, there is currently no complete guideline on the advantages, disadvantages, and usability of various algorithms^[33]. This has led to the disconnection of research in this field, and the progress of practice has not been shared in time, resulting in the slow progress of boiler combustion optimization. This paper systematically summarizes the current achievements of boiler combustion optimization, aiming to provide guidance for interested researchers.

The use of intelligent optimization technology in boiler combustion optimization is mainly divided into two stages, namely, the establishment of the boiler combustion model and the optimization of the combustion process. Using the historical data of boiler combustion, the model is established by EC and DM technology, and then the combustion process is optimized by EC technology. The optimization flow chart is shown in Fig. 3.

Table 1 Works summar	y of mechanism methods.
----------------------	-------------------------

· .	
Category	Reference
Physical and chemical methods	[5, 7–14]
Numerical simulation	[6, 15–24]
Burner design optimization	[25-30]



Fig. 3 Computational intelligence technologies to optimize boiler combustion processes.



Fig. 4 Number of EC technical works in boiler combustion optimization.

Figure 4 shows the application of the main popular EC technologies in boiler combustion optimization. It can be clearly seen from Fig. 4 that genetic algorithms (GAs) and particle swarm optimization (PSOs) have relatively more research, while ant colony optimization (ACOs), teaching and learning-based optimization (TLBOs), differential evolution (DEs), and artificial bee colony (ABCs) are relatively few. The number of other EC technologies is small, mainly including grey wolf optimization (GWOs), whale optimization algorithm (WOAs), vortex search (VSs), and so on. In addition, the works on single-objective optimization are more than that on multi-objective optimization. (Please note that the data may not be absolutely complete, but are representative works. The data are from mainstream databases such as Google Scholar and Web of Science.) The goal of this paper is to conduct a comprehensive investigation of boiler combustion optimization research, hoping to further promote the research in the field of boiler combustion optimization.

The remaining sections are as follows. Section 1 introduces the structure and mechanism of CFBB. Section 2 introduces the existing researches on boiler combustion optimization. Section 3 presents future research directions and challenges. Section 4 summarizes the paper.

1 Background

This section introduces the basic structure and characteristics of boilers, the principle of NO_x generation, and the calculation method of thermal efficiency.

1.1 CFBB structure and combustion process

Figure 5 shows the structure of a CFBB boiler in a thermal power plant. The coal is first broken into small particles, and then sent into the furnace together with limestone and circulated under the action of primary and secondary air. To achieve full utilization of raw materials, the cyclone separator recycles unburned pulverized coal particles into the furnace. The high temperature and highpressure gas produced by combustion will enter the turbine, driving the generator to generate electricity and converting heat and mechanical energy into electrical energy. Finally, the flue gas is filtered by the bag filter and discharged into the atmosphere.

Pulverized coal particles in a CFBB are fluidized under the action of graded wind, which is a unique feature different from the traditional pulverized coal boiler, and its unique characteristics have many advantages.

(1) It can adapt to various fuel types with a high fuel utilization rate. Some inferior fuels with low calorific value, high ash content, and low ash melting points, such as lignite and peat, can also be entirely burned in CFBB.

(2) In actual production, the CFBB bed temperature is strictly



Fig. 5 Structure diagram of CFBB.

maintained at 850–950 $^{\circ}$ C, lower furnace temperature, effectively avoiding the slag melting phenomenon. By adding limestone to the combustion process, the ash contains high calcium content, which can be used to make raw materials for cement and make the combustion residue recycled.

(3) The boiler has a wide range of manipulation conditions. This type of boiler load has an extensive mobilization range, which can meet the complex needs of the State Grid in different seasons and different periods.

(4) CFBB helps reduce pollutant emissions. The low-temperature combustion environment of CFBB makes it difficult for nitrogen (N) in the air to oxidize to NO_x. In addition, the staged combustion process can broadly inhibit the conversion of N into NO_x in the fuel and reduce some of the NO_x generated. Therefore, the combustion of CFBB is a kind of energy-saving and environmental protection combustion mode.

1.2 NO_x formation mechanism

 NO_x is the main pollution gas in the CFBB combustion process, mainly including NO and NO_2 , and NO accounts for about 90%. In order to prevent air pollution and protect the environment, it is necessary to strictly control its emission standards. CFBB combustion is a very complex process of physical and chemical changes. According to the NO_x formation mode, it can be roughly divided into three types: fuel type, thermal type, and fast type.

(1) Fuel NO_x is generated during the combustion process and is the dominant form. After a series of complex physical and chemical changes, numerous nitrogen ions and nitrogencontaining compounds will be decomposed. A series of chemical reactions occur between high-temperature conditions and carbon, hydrogen, and oxygen plasma, and NO_x is generated. This is also the main source of NO_x generated in the operation of CFBB, accounting for about 90% of the total NO_x production. The redox process of nitrogen in fuel is shown in Fig. 6.

Studies have shown that NO_x generated by coal combustion mainly comes from two ways, volatile ammonia and coke ammonia. In the early stage of the eruption, volatile coal ammonia produces some intermediate products in the rich zone. If oxygen is sufficient, NO_x is produced. If oxygen is insufficient, the precipitated nitrogen oxides are reduced to N_2 . Because fuel NO_x accounts for a large proportion of total NO_x emissions, it is very important to study its formation and destruction mechanisms to control NO_x emissions.

(2) Thermal NO_x comes from the oxidation reaction of N₂ in the air at high temperature. The primary air and secondary air in the furnace are rich in N₂ and O₂. In a high-temperature environment, a series of chemical reactions occur between N₂ and O₂ to generate NO_x. The production process is shown in the following equation:

$$O + N_2 \rightarrow NO + N$$
 (1)

$$N + O_2 \rightarrow NO + O$$
 (2)

The production of thermal NO_x is greatly affected by temperature. If the temperature is higher than 1200 °C, its production will be large. On the contrary, if the temperature is lower than 1200 °C, its production is very small. Since the



temperature in the CFBB furnace is generally maintained at 850–900 °C, the thermal NO_x production accounts for a small proportion of the CFBB.

(3) Fast NO_x is produced by the reaction of hydrocarbon ion clusters in fuel and N in the air according to the Fenimore mechanism. Fast NO_x production requires more hydrocarbons, and the oxygen concentration is relatively low. The production of rapid NO_x generally does not exceed 5% of the total NO_x production, so the production of rapid NO_x is less than that of thermal NO_x. Usually, only with gas-fueled burners can their production be considered.

1.3 Thermal efficiency calculation

Most of the heat generated during the regular operation of the boiler will be absorbed through the water wall. The absorbed heat is usually called adequate heat, and the part of heat lost through other ways is called heat loss. The thermal efficiency of a boiler is the percentage of heat energy converted to the total heat energy input. Thermal efficiency is not only the standard to evaluate the operation status of the boiler, but also an essential goal of boiler combustion optimization. The calculation method is shown as fouows:

$$\eta = \frac{Q_1}{Q_r} \times 100\% \tag{3}$$

where η is the thermal efficiency of the boiler, Q_1 is the effective heat, and Q_r is the total heat input.

(1) Calculation method of thermal efficiency. There are two methods to calculate thermal efficiency, the positive equilibrium method and the inverse equilibrium method. The positive balance method is to obtain Q_1 and Q_r through measurement and then calculate according to Eq. (3), namely, the thermal efficiency of the boiler. The inverse equilibrium method is also known as the heat loss method because it measures the heat loss value of the boiler combustion system and then calculates the percentage of total heat loss to Q_r , as well as the combustion efficiency. The calculation equation is as follows:

$$\eta = 1 - \frac{Q_s}{Q_r} \times 100\% = 1 - \sum q \tag{4}$$

where Q_s is the sum of all boiler heat loss.

Many thermal power plants obtain relevant data by measuring the speed of the coal feeder or using belt weighing and other rough means, so the measured data must have mistakes, resulting in within errors the calculated thermal efficiency. The equilibrium and anti-equilibrium methods are the main methods for calculating the thermal efficiency of boilers. Suppose the relative error is δ when the total heat input is Q_r , we have:

$$\Delta_{1} = \frac{Q_{1}}{Q_{r}(1\pm\delta)} - \frac{Q_{1}}{Q_{r}} = \frac{\pm\delta\cdot Q_{1}}{(1\pm\delta)\cdot Q_{r}}$$
(5)

$$\Delta_2 = \left(1 - \frac{Q_s}{Q_r(1 \pm \delta)}\right) - \left(1 - \frac{Q_s}{Q_r}\right) = \frac{\pm \delta \cdot Q_s}{(1 \pm \delta) \cdot Q_r}$$
(6)

It can be found from Eqs. (5) and (6) that the difference between the error values Δ_1 and Δ_2 of the positive and negative balance methods is determined by the size of Q_1 and Q_s , respectively. The combustion efficiency of CFBB is relatively high. Generally, Q_1/Q_r is about 90%, and Q_s/Q_r is about 10%. It can be seen that the error Δ_2 obtained by the inverse equilibrium method is about 1/9 of that obtained by the positive equilibrium method Δ_1 . (2) Anti-equilibrium calculation method. Limestone is usually added as a desulfurized in the CFBB combustion process. The addition of limestone will affect the change of material composition and energy in the furnace, which is not conducive to calculating thermal efficiency. For the convenience of mathematics, the m_i kg desulfurized required for each kilogram of coal combustion is taken as the standard, and the changes of material and heat in the furnace brought about by the combustion process are equivalent to the corresponding mass of coal. The anti-equilibrium calculation Eq. (4) is transformed into Eq. (7).

$$\eta = 1 - \frac{Q_2 + Q_3 + Q_4 + Q_5 + Q_6 + Q_7}{Q_r} \times 100\% =$$

$$1 - (q_2 + q_3 + q_4 + q_5 + q_6 + q_7)$$
(7)

where Q_2 is the physical heat loss of exhaust smoke (kJ/kg). Q_3 is incomplete combustion heat loss of combustible gas (kJ/kg). Q_4 is solid incomplete combustion heat loss (kJ/kg). Q_5 is the boiler heat loss (kJ/kg). Q_6 is the physical heat loss of ash (kJ/kg). Q_7 is the heat loss of desulfurization reaction (kJ/kg), $q_i(i = 2, 3, ..., 7)$ is the Q_i percentage of Q_r .

It can be seen from Eq. (7) that when the inverse equilibrium method is used to calculate the thermal efficiency, the calculation results of each heat loss will affect the thermal efficiency of the boiler. Therefore, in order to obtain a more accurate thermal efficiency of the boiler, it is necessary to accurately calculate the heat loss of the boiler, to lay the foundation for the combustion optimization of each index in Eq. (7) can be referred to in Ref. [34]. Among the six heat losses, exhaust heat loss Q_2 is the largest proportion of the heat loss Q_4 caused by incomplete combustion of solids is second only to exhaust heat loss, and the proportion is also large. The ash physical heat loss Q_6 has a greater impact on the index is small.

2 Existing Researches in Boiler Intelligent Optimization

This section summarizes the existing researches on boiler combustion optimization. It can be seen from Fig. 7 that according to different technologies, the works of boiler combustion optimization are divided into two categories, namely the EC paradigm and the DM paradigm. Mainstream EC technologies have been used in boiler combustion optimization. Genetic algorithm is a typical evolutionary algorithm, particle swarm optimization, ant colony optimization, and teaching and learning-based optimization are representative swarm intelligence algorithms. Other algorithms include differential evolution and vortex search, which are relatively lacking. DM paradigm is mainly used in boiler combustion modeling, including artificial neural networks (ANNs) and support vector machine regression (SVRs). Some other machine learning techniques such as ensemble learning and deep learning have less work.

2.1 EC paradigms

This subsection summarizes the works of EC technologies in boiler combustion optimization. It mainly includes GAs, PSOs, ACOs, TLBOs, and other EC techniques.

2.1.1 GAs for boiler combustion optimization

GAs are likely to be the first intelligent algorithm for boiler



Fig. 7 Existing researches classification of intelligent boiler combustion optimization.

Table 2 Representative works of GAs in boiler combustion optimization.

Number of objectives	Reference
Single-objective	[35-49]
Multi-objective	[50-54]

combustion optimization. Since the 1990s, GAs have been applied to the optimization of boiler thermal efficiency. Table 2 lists some representative works on GAs optimization of boiler combustion. It can be seen in Table 2 that the number of single-objective optimization is more than that of multi-objective optimization.

In order to estimate the oxygen content in syngas generated by biomass in different types of facilities, Krzywanski et al.[55] considered the optimization of calcium oxide adsorption in syngas by bubble fluidized bed (FB) and CFBB. A non-iterative model was developed using GA and ANN to optimize hydrogen production. The experimental results showed that based on the established model, the thermal efficiency of CFBB was improved. The maximum hydrogen production rate was 67.4%, the temperature reached 775 °C, and the molar ratios of CaO/C and H₂O/C were 2.40 and 3.12, respectively. Suresh et al.^[56] used ANN and GA to study the thermal efficiency optimization of a high-ash coal-fired boiler in a supercritical power plant in order to maximize the thermal efficiency of the boiler. Wang et al.^[57] have realized a safe and efficient explosion of boilers to some extent by using fuzzy association mining algorithms, ANN, and ECs in computational intelligence technologies. In Ref. [58], the established ANN-GA model was verified by historical combustion data and computational fluid dynamics. The experimental results showed that computational fluid dynamics was helpful to improve the performance of ANN and provided an effective tool for the combustion optimization of CFBB in practical industrial operations.

There are also many research works in multi-objective optimization. Chaudhari and Garg^[90] used single-objective and multi-objective optimization techniques to study the adjustment model of maleic anhydride CFBB commercial plant reactor. Using the NSGA-II algorithm, the optimal productivity parameter combination was obtained. Buche et al.^[60] proposed a multi-objective evolutionary algorithm capable of dealing with noise problems, which was applied to boiler combustion optimization to reduce the emission of NO_x and the pressure fluctuation of flame. In Ref. [61], based on the experimental data of coal-fired boilers, the hybrid model for predicting the efficiency and pollutant emissions of 360 MW boilers was established by using ANN. The GA was used to reduce the fuel and environmental costs as the

optimization objective, and the optimal operation schemes such as excess air, primary airflow, and secondary air flow were obtained. This algorithm was combined with distributed control system (DCS) to realize the real-time coordinated and optimized control of power plant boilers. Shi et al.^[6] used the computational fluid dynamics simulation method combined with ANN, the thermal efficiency and NO_x emission models of a 660 MW ultra-supercritical boiler were established, and the average prediction errors were 0.04% and 3.56 mg/Nm³, respectively. Then the GA was used to optimize the air supply scheme to achieve the goal of high thermal efficiency and low NO_x emission.

In general, GA has played an excellent optimization effect in boiler combustion. However, GA still faces many challenges in boiler combustion optimization, such as difficulty in selecting hyperparameters, slow convergence speed, and easily falling into local minima.

2.1.2 PSOs for boiler combustion optimization

Table 3 lists the representative works of PSOs in boiler optimization. It can be seen in Table 3 that there are more studies on single-objective optimization than multi-objective optimization.

Because the original PSO has a fast convergence speed and is easy to converge to the local optimum, it is difficult to achieve a better optimization effect. Many scholars have used improved PSO combined with regression models such as ANN and SVR for boiler combustion optimization. In order to make full use of the potential of CFBB waste heat sources, Garg and Orosz^[86] used PSO to search for the best trade-off between specific investment costs and energy utilization. A method for generating high entropy efficiency vortex geometry corresponding to the optimization period was proposed, and the optimization analysis was further extended to solar thermal applications. In Ref. [63], it focused on the use of PSO and SVR to minimize NO_x emissions and compared the performance with GA and ACO algorithms in this research. In Ref. [87], PSO with variable population size was used to optimize the reliability of the boiler furnace system. The results showed that the optimized reliability was as high as 99.9845%. In Ref. [88], a new hybrid jump particle swarm optimization

Table 3	Representative works of PSOs in	boiler combustion optimization.
---------	---------------------------------	---------------------------------

Number of objectives	Reference
Single-objective	[63-80]
Multi-objective	[81-85]

(HJPSO) was proposed to adjust the gain of the boiler proportional-integral (PI) controller. After introducing Gauss and Cauchy mutation, the optimization ability of HJPSO was greatly enhanced, and better PI controller gain was obtained in the comparison of other PSO variants. In Ref. [89], the distributed PSO based on MapReduce was used to optimize the thermal efficiency and NO_x emission of the boiler. The weighted coefficient method was used to convert multi-objective optimization into single-objective optimization. In Ref. [90], the control effects of PSO-proportional-integral-derivative (PID), modern controller fuzzy logic controller (FLC), and classical PID controller on CFBB bed temperature were compared. The results showed that the control establishment time of PSO-PID was shorter than that of other controllers.

In general, the application of PSO in boiler optimization is reflected in the improvement of PSO, and the optimization performance of PSO is improved by designing different strategies. However, most of the works do not consider the characteristics of boiler combustion to improve the algorithm.

2.1.3 ACOs for boiler combustion optimization

Table 4 lists the works of ACOs in boiler combustion optimization, the earliest work can be traced back to 2008^[91]. There are not many studies on ACOs in boiler combustion compared with GAs and PSOs. In addition, it can be seen from the statistical data that more work focuses on single-objective optimization, and there are very few works on multi-objective optimization.

In Ref. [106], SVR was used to establish a model of the relationship between boiler NO_x emission concentration and boiler operating parameters. Then ACO was used to optimize the NO_x emission concentration. The experimental results showed that in the comparison of GA and PSO, the decision scheme optimized by ACO can effectively reduce NO_x emissions. In Ref. [107], a novel ACO-SVR model was proposed to establish a NO_x emission model. Experimental results showed that the proposed algorithm has higher prediction accuracy and training speed. Zheng et al.^[109] introduced the combination of SVR and ACO to reduce NO_x emissions. Experiments showed that the method used effectively reduces NO_x by about 18.69%.

ACO has relatively less work in boiler combustion optimization. ACO has the characteristics of fast global convergence and strong robustness, but it is prone to stagnation and falls into local optimum in the later stage of evolution. Therefore, it is still necessary to study and design effective strategies to enhance the optimization efficiency of ACO in boiler optimization. In addition, it is necessary to further explore multiobjective optimization technology to meet the requirements of boiler optimization development.

2.1.4 TLBO for boiler combustion optimization

Table 4 Representative works of ACOs in boiler combustion optimization.

Single-objective	[91–104]
Multi-objective	[105]
Table 5 Representative works of optimization	TLBOs in boiler combustior
Table 5 Representative works of optimiza	TLBOs in boiler combustion tion.
Table 5 Representative works of optimiza Number of objectives Single-objective	TLBOs in boiler combustion tion. Reference [38, 80, 110–118]

As an efficient swarm-based search algorithm, TLBO is widely used in various industrial problems^[109]. Table 5 shows the works of TLBO on boiler combustion optimization. In order to improve the optimization efficiency of boiler combustion, more improved TLBOs have been proposed.

In Ref. [122], a novel teaching-learning-based optimization algorithm (TLSO) was proposed, which dynamically adjusted the convergence speed according to the number of iterations and enhanced the global search ability of the algorithm. Compared with several other algorithms, such as PSO, ACO, TLBO, etc., the experimental results showed that TLSO achieves better results in NO_x emission optimization. Li et al.^[111] proposed an improved A-TLBO algorithm to construct a 330 MW CFBB combustion model. The experimental results showed that the proposed method had good regression accuracy and generalization ability. In Ref. [123], the improved AELM model was combined with TLBO to establish a CFBB combustion NO_x model. The experimental results showed that compared with the other six models, AELM-TLBO had better prediction accuracy and provides a basis for reducing NO_x emissions. In Ref. [121], a multiobjective improved teaching optimization algorithm (MMTLBO) was proposed. Then, the thermal efficiency, NO_{y_2} and SO_{y_3} in the boiler combustion process were comprehensively optimized. The experimental results showed that MMTLBO can find multiple sets of reasonable combustion schemes and provide meaningful guidance. In Ref. [114], an improved TLBO algorithm was used to optimize the NO_x emission of boiler combustion. The experimental results showed that the proposed method can effectively reduce the NO_x emission concentration. In Ref. [124], an improved TLBO algorithm (MTLBO) was proposed. The optimization ability and generalization performance of MTLBO were verified by 14 standard test functions. Finally, the NO_x emission of a 330 MW boiler was optimized. The experimental results showed that the proposed MTLBO algorithm has good optimization performance in NO_x emission optimization.

As an efficient optimization algorithm, the main advantage of TLBO is that it does not require parameter setting. Unlike PSO and GA, the optimization performance is greatly affected by algorithm parameters. However, the computational complexity of TLBO in boiler combustion optimization is relatively high, and it is necessary to further improve the computational efficiency to meet the requirements of complex and variable working conditions.

2.1.5 Others EC techniques for boiler combustion optimization

Table 6 lists other optimization techniques in boiler combustion optimization, including differential evolution (DE), artificial bee

Algorithm	Reference
DE	[70, 125–130]
ABC	[38, 131–134]
WOA	[135–137]
GWO	[138-140]
VS	[141, 142]
HS	[143]
JAYA	[144]
SSA	[145]

colony (ABC), whale optimization algorithm (WOA), sparrow search algorithm (SSA), grey wolf optimization (GWO), vortex search (VS), harmony search (HS), and some mathematical optimization techniques. Among them, DE has the most research results. In addition, Refs. [130, 131] are multi-objective optimization works.

Pattanayak et al.^[166] developed a groundbreaking optimization system based on thermodynamics and an ANN model. The system used an optimization algorithm to refine the search for the optimal sequence of fan frequencies, which improved thermal efficiency and reduced pollutant emissions. In Ref. [147], a novel evolutionary multi-objective search algorithm was used to find the probability trade-off front between NO_x and fly ash carbon content based on a data-driven model, which effectively solves the problems of NO_x and thermal efficiency. In Ref. [148], the boiler efficiency was optimized using the optimal manipulated variable (MVs) decision algorithm.

In Ref. [149], multivariate analysis tools such as principal component analysis (PCA) and partial least square-regression (PLS-R) were used to establish the relationship between the input parameters of biomass gasification in a CFBB furnace. According to the optimization results, in order to obtain high quality, it was suggested to use olivine with a high carbon conversion rate and a low tar yield. Kusiak and Song^[150] used a data mining method to optimize the complex, nonlinear, and non-stationary combustion process of CFBB. On this basis, a virtual test program was developed to verify the results of the optimization method. The results showed that the developed program improved combustion efficiency. Kim et al.^[151] pointed out that the input parameters are related to the performance of the established model. When selecting the input parameters, not only the accuracy of prediction should be improved, but also the dimension of the model should be reduced. Therefore, one or more input parameters should be eliminated continuously. No less than 10 input parameters were saved out of the 36 initial input parameters. Finally, through the test of four commonly used models, it was found that the least squares support vector machine (LSSVM) combined with principal component analysis had the smallest prediction error of NO_x emission, which laid the foundation for boiler combustion optimization. In Ref. [152], a NOx emission prediction model based on the combination of a feature selection method using an improved pollination algorithm and a random forest was proposed. The results showed that the model has high prediction accuracy and good robustness, which provides key technical support for combustion optimization. A model framework based on deep belief network (DBN) and JAYA was developed to obtain a 660 MW CFBB combustion model and optimize NO_x emission concentration^[144]. In Ref. [138], GWO was used to optimize the CFBB operating parameters and improved the operating efficiency. In the comparison between GA and PSO, the proposed method effectively reduced the calculation time and improved the solution accuracy.

Although various intelligent optimization methods have promoted the research of boiler combustion optimization, the proposed algorithms are rarely aimed at boiler combustion characteristics, and the efficiency of the algorithm needs to be further improved.

2.1.6 Performance summary of EC technologies

Boiler combustion optimization refers to improving combustion efficiency and economy and reducing environmental pollution by adjusting boiler combustion parameters. As an effective tool to find the optimal solution, intelligent optimization algorithms have been widely used in boiler combustion optimization. This subsection discusses the effects of various EC algorithms in boiler combustion optimization and related connections and differences.

In boiler combustion optimization, GAs can optimize and adjust the adjustable parameters by modeling the fitness function of combustion parameters. GAs can solve multi-objective optimization problems and nonlinear problems to a certain extent, but for problems with a large distribution of decision space, it may lead to reduced search efficiency. PSOs find the optimal solution by simulating the movement and interaction of particles in the search space. PSOs have good search ability for problems with large search space, but for non-convex problems, they may fall into local optimal solutions and cannot obtain a significantly improved decision scheme. ACOs is a search algorithm based on the behavior of ants in finding food. In boiler combustion optimization, ACOs can find the optimal boiler combustion operating conditions by simulating the pheromone deposition and volatilization of ants during the search process. ACOs can deal with multi-objective optimization problems and constrained optimization problems, but they can not play a good optimization effect on problems with large search spaces. TLBOs find the optimal solution by simulating the behavior of teachers and students in the learning process. TLBOs can also handle multiobjective optimization problems and constrained optimization problems. TLBOs have good optimization ability in the early stage of the search stage, but the convergence speed is relatively slow in the later stage of the search.

Among the above four optimization algorithms, GAs, and PSOs are widely improved and applied to boiler combustion optimization. Due to the difference in the search performance of each algorithm, it can produce different optimization effects under different working conditions. In future work, it is necessary to further analyze the advantages and disadvantages of each algorithm in order to achieve better search performance in boiler combustion optimization.

2.2 DM paradigms

This subsection summarizes the works of DM algorithms in boiler combustion optimization. It mainly includes ANNs, SVRs, and other DM algorithms.

2.2.1 ANNs

Artificial neural networks (ANNs) have been proven to be an effective modeling method and have been widely applied to various fields of the power generation industry^[153], boiler failure detection^[154-156], noise anomalies^[157], power output prediction^[158, 159], and early abnormal behavior changes^[160, 161]. This helps give untimely warnings during operations^[162]. When some studies focus on monitoring and control^[166-167], others investigate the possibility of optimization^[166-170] and establish intelligent power plants^[171]. Since the performance of ANNs in CFBB optimization was clarified in the 1990s^[172], a large number of studies using ANNs applications have been produced. Table 7 shows the application of some

Table 7 Representative works of ANNs variants in boiler combustion optimization.

Algorithm	Reference
BP	[37, 68]
ELM	[159, 173–175]
DT	[157]
LSTM	[158, 176]

representative ANNs variants in boiler combustion optimization, including back propagation (BP), extreme learning machine (ELM), decision tree (DT), long short-term memory recurrent neural network (LSTM).

In Ref. [174], in order to improve the performance of ELM, an improved extreme learning machine was proposed. A new method was adopted to generate hidden layer input weights and deviations, and a new hidden layer activation function was presented. In order to verify the validity of the proposed model, a 330 MW boiler combustion process was modeled and good results were obtained. Liang et al.^[72] established a multi-objective model of boiler combustion using an optimized extreme learning machine network to predict boiler thermal efficiency and NO_x emissions. In Ref. [177], a scheme for boiler combustion in functional power plants was proposed. In this study, the ANN and LSSVM were used to conduct Monte Carlo experiments and interval adjoint dominance analysis, which reduced some operating variables. The experimental results showed that the efficient generation of ANNs under the unit load of a power plant was obviously more effective. In Ref. [178], a CFBB gasifier/steam turbine/proton exchange membrane (PEM) fuel cell integrated system was developed. The system contains manv thermochemical, biochemical, and physical processes. The ANN was established after combining the reaction principle and CFBB gasifier operating parameters. This model could accurately predict the output parameters of the PEM fuel cell and provide corresponding guidance. In Ref. [179], in order to predict the bed temperature of CFBB more accurately, after analyzing the chemical combustion process of coal-fired units, it is considered that different coal types have a great impact on the prediction of bed temperature. In order to solve this problem, the coal type information was added to the initial deep neural network (DNN) model. In the experiments on two real CFBBs, the proposed model accurately predicted the bed temperature. Booth and Roland^[180] optimized boiler operation by using online learning neural networks, adjusted fuel quality fluctuations, and improved operational flexibility according to equipment performance changes caused by wear and maintenance activities. In addition, some boiler combustion optimization techniques have been applied at power plant sites. For example, Ultramar developed boiler combustion optimization software using statistical methods. The smart process of emerson company (SPEC) was based on ANN, used optimization techniques to optimize the boiler combustion process, and was successfully applied to the Warsaw Ostroleka power plant^[181]. Pegasus has designed and developed the NeuSIGHT system and the PowerPerfect system. Using the data from DCS, the optimal control of boiler combustion is realized through the ANN^[182]. Zhou et al.^[183] established an ANN for low NO_x combustion performance of high-capacity boilers by using the nonlinear dynamic characteristics and self-learning characteristics of ANN. The model can predict the NO_x emission concentration and unburned carbon content under different working conditions. Combined with the optimization algorithm, the optimal operating conditions for low NO_x combustion were found. In Ref. [143], the improved ELM model was used to optimize the operation parameters of a 700 MW CFBB to reduce NO_x emissions. In addition, the proposed algorithm was compared with several other advanced algorithms to obtain better results and stronger robustness. The thermal efficiency of a 600 WM boiler was predicted using the ANN model, and the results demonstrated that the proposed method was more accurate in predicting the thermal efficiency of the boiler and other indicators. Finally, the model was tested using statistical

methods^[184].

2.2.2 SVRs

References [185-188] used different artificial intelligence optimization algorithms to optimize the hyper-parameters in support vector machines to obtain the optimal model of boiler combustion and then optimized the obtained model under multiparameter conditions by using the optimization algorithm to achieve the goal of reducing the emission of boiler pollutants and improving the thermal efficiency of the boiler. An online combustion optimization system was designed to reduce the NO_x concentration generated by combustion. SVR and random forest models were used to verify the neural network model^[69]. In order to meet the needs of bed temperature prediction, a dynamic model for predicting 300 MW CFBB by using LSSVM and delayed order group optimization technology^[74]. On the basis of feature selection, the ensemble learning model was applied to decompose and combine the data subset based on LSSVM to predict NO_x emissions^[189]. In Ref. [190], the adaptive LSSVM algorithm was used to establish the CFBB combustion model, and a two-step learning strategy was designed and verified by a nonlinear model. In Ref. [191], the Gaussian process model was used to establish the CFBB combustion model, and the SVM model was compared and verified.

2.2.3 Others DM techniques

In Ref. [192], the ensemble learning method was applied to CFBB historical combustion data to predict the unit. Random forest prediction results showed that radial position has the greatest impact on local mass flux and species separation. The overall mass flux had the greatest impact on local particle concentration, while the impact on local clustering characteristics was relatively weak. In addition, if large enough datasets were used, the ensemble learning model could be trained as a model with good predictive ability without any understanding of how it works. This study emphasized the value of machine learning methods in CFBB optimization. In Ref. [193], a NO_x emission prediction method based on the stacking generalized ensemble method (SGEM) was proposed. The PCA method was combined to select the relevant variables. In SGEM, the ensemble learning model was constructed with a back propagation neural network (BPNN), SVR, and DT as the basic models, and linear regression (LR) as the meta-model. Through experimental verification, the proposed method has high accuracy in predicting NO_x emissions. In order to predict the thermal efficiency and NO_x emission concentration of CFBB boilers, a deep bidirectional learning machine was proposed and compared with the same type of ELM^[175]. In order to balance boiler thermal efficiency and emission control, a reinforcement learning architecture based on LSTM and deep Q-network structure was developed^[176].

3 Future Direction and Research Challenge

With the continuous development of computer technology, many foreign companies have launched several boiler combustion optimization systems. In the 1990s, Pegasus Corporation of the United States developed the first NeuSIGHT system based on neural network modeling for power systems, which worked well in the capacity efficiency of coal-fired power plants at the time^[194]. Based on this system, by adding a computer off-line simulation function and dynamic prediction function, the company has developed a Power Perfecter system with even better performance. This system can reduce NO_x emission by 10%–30% while

improving boiler thermal efficiency by 0.5%-2.5%, and the optimization effect is very satisfactory. Ultramax has introduced the Ultramax system, which has been successfully used in more than 30 countries and regions^[195]. Using multiple optimization algorithms to model artificial neural network parameters optimally, Emerson eventually developed a SmartProcess system that has improved thermal efficiency by approximately 0.6% and reduced NO_x emissions by approximately 13% for the 200 MW coal-fired unit at the Ostroleka power plant in Warsaw^[196]. The GNOCISPLUS combustion optimization system, developed by Powergen based on historical operating data of power stations, has been applied in 37 power stations in the UK and the US. The system uses a neural network modeling approach as a tool to control emissions of NO_x and other pollutants by combining it with the plant's intelligent control system. This system uses neural network modeling as a tool to control the emission of polluting gases such as NO_x by combining it with the intelligent control system of the power plant. In addition, many DCS manufacturers, such as Honeywell, Siemens, Emerson, etc., have added combustion optimization control strategies to their mature control system products.

Although there are many combustion optimization systems available, the algorithms and techniques they use are relatively simple. On the other hand, these systems have made good impact abroad. However, China is a vast country with diverse coal quality conditions and a very complex combustion environments. Therefore, it is difficult for foreign systems to play a good role in guiding optimization in China. Further research and design of combustion optimization systems are needed in China^[197]. Some of the mainstream research directions and challenges are listed below.

3.1 Combustion data processing

Most power plants are currently using DCS systems or SIS systems to upgrade their information technology, in which a large amount of historical boiler operation data is stored in the DCS system or SIS system. Although these massive data help to model the boiler combustion system, the boiler operation site is complex and the information collected by different testing devices is more chaotic and disorderly, which requires the use of effective data mining methods to process these system data and then select effective data to build the boiler combustion model.

Based on a data-driven model, Ref. [198] monitors the actual boiler efficiency and its desired efficiency. The method was based on an information-theoretic variable ordering of the massive data. In the data processing of ultra-supercritical units, the model structure is a transfer function matrix through dynamic analysis, and the data-driven multivariate model parameters are proposed in the form of intelligent identification combined with evolutionary algorithms for data mining, and the work done further provides a reference for boiler control^[199].

For data models, the quality of the dataset is very important. It is important to extract more effective information from the large amount of data obtained from DCS for boiler combustion optimization.

3.2 Feature selection and feature reconstruction

The boiler combustion system is a nonlinear, strongly coupled, large hysteresis system. Its operating parameters are numerous and have complex coupling relationships with each other. If many parameters are directly used as inputs to the black-box model to construct the boiler combustion system model, it will inevitably make the input information of the model redundant and thus weaken the generalization ability of the model^[200]. Therefore, when constructing the boiler combustion model, it is necessary to analyze the correlation between the operating parameters that affect the combustion characteristics of the boiler and then eliminate the correlation between the variables.

This study is divided into two main categories. One category is the use of feature extraction. Reference [201] uses a deep Q network (DQN) and an LSTM module to extract features of the combustion process from the DCS operating data. The other category is the use of feature reconstruction. Reference [92] used the mutual information (MI) algorithm to calculate the importance of the actual variables, analyzed them to determine the variables of interest, and then reconstructed the modeled data.

After using the feature selection technique for boiler parameters, the performance of the data model was improved^[202, 203], and further research on the technique of boiler feature selection is important to improve boiler combustion optimization.

3.3 Boiler combustion modeling

Advanced computational intelligence techniques are introduced into boiler combustion optimization. Based on historical data of boiler combustion, combustion models are built using machine learning methods, such as neural networks^[204], support vector machines^[202, 205, 206], random forests^[207, 208], etc. On the established model, the thermal efficiency and pollutant emission concentration of the boiler are optimized using population intelligence optimization techniques to achieve safe and efficient low-pollution combustion of the boiler.

These algorithms are capable of modeling the data of boiler combustion and have great advantages in modeling complex systems. It can extract the mapping relationship between the sample input and the target output by learning from a large amount of sample data. However, the traditional neural network relies too much on the random initial values of the network and also suffers from the problems of difficulty to determine the network model and slow learning speed. The parameters of support vector machines also rely too much on empirical knowledge. Therefore, its application in boiler combustion optimization needs to be further improved and perfected.

3.4 Knowledge-driven optimization

The data obtained from DCS are plagued by uncertainties and deficiencies. Hybrid prediction techniques for process control systems are the norm today and involve a combination of datadriven and knowledge-driven models.

Reference [209] generated an artificial neural network prediction tool using Visual Basic GUI and combined it with expert knowledge for predicting the spray values of a 500 MW boiler within the allowed tolerances. Reference [210] considers the structure of distributed support systems and proposes an integration framework based on a distributed architecture that includes expert knowledge and collaboration techniques, tools that work with decision-makers to enrich their knowledge.

The combination of expert knowledge and problem knowledge guides the combustion optimization of the boiler to good effect. It is of good practical importance to further study the combination approach.

3.5 Multi-objective optimization

Most of the current boiler combustion optimization methods and techniques are only optimized for a single objective, just optimizing boiler NO_x emissions or thermal efficiency. There are relatively few studies that simultaneously perform multi-objective

integrated optimization of both. The existing algorithms mainly include immune cell subpopulation multi-objective optimization algorithm^[21], multi-objective bee colony algorithm^[22], multiobjective teaching and learning optimization algorithm^[121], etc. However, these methods are not designed with the characteristics of combustion problems, and the comprehensive optimization capability needs further improvement.

3.6 Low carbon policy

To achieve the target task of carbon peaking and carbon neutrality, China has increased the efforts of boiler energy saving and emission reduction, and implemented a series of policy measures such as coal power structure optimization and transformation and upgrading, coal-fired industrial boiler energy saving and environmental protection comprehensive upgrading project, etc. The level of boiler energy saving and environmental protection has been significantly improved.

While reducing NO₂, SO₂, and other polluting gases, CO₂ emissions need to be further reduced. For example, CO₂ has been reduced by a new method of using flue gas heat in cascade^[213]. A process simulation of a conventional plant was set up in Ref. [214] using Aspen plus, resulting in a CO₂ recovery rate of 96.24% for the whole system. In the future, the CO₂ emission concentration can be further reduced in terms of the structural design of power generation boilers^[216] and optimization of various parameters^[216].

4 Conclusion

This paper provided a complete investigation of the applications of intelligent optimization algorithms in boiler combustion optimization. It mainly included the commonly used EC algorithms and analyzed the advantages and disadvantages of each algorithm. In addition, the problems and challenges in future boiler combustion optimization were pointed out and discussed.

The survey shows that EC technologies are widely used in the field of boiler combustion optimization because of their high search efficiency. Each algorithm has its characteristics, e.g., GAs can retain genetic characteristics in the process of evolution, the update mechanism of PSOs is simple, the computational cost is relatively small, TLBOs have high global search ability, and so on. Therefore, the improved versions of these EC technologies can better optimize the boiler combustion process and deserve further study.

Although the current stage of works has achieved good optimization results, there are still many issues that need further in-depth research, such as, pre-processing of boiler combustion data, feature selection and feature reconstruction of combustion process parameters, numerical and data-based modeling, using knowledge-driven assisted optimization, and meeting low-carbon policy requirements. In addition, the algorithms need to have certain generalization performance as the corresponding policies are developed.

Acknowledgment

This work was supported by the National Natural Science Foundation of China (Nos. 61806179, 61876169, 61922072, 61976237, 61673404, 62106230, 62006069, 62206255, and 62203332), China Postdoctoral Science Foundation (Nos. 2021T140616, 2021M692920, 2022M712878, and 2022TQ0298), Key R&D Projects of Ministry of Science and Technology (No. 2022YFD2001200), Key R&D and Promotion Projects in Henan Province (Nos. 192102210098 and 212102210510), and Henan Postdoctoral Foundation (No. 202003019).

Article History

Received: 19 December 2022; Revised: 23 February 2023; Accepted: 22 March 2023

References

- J. S. Gaffney and N. A. Marley, The impacts of combustion emissions on air quality and climate – From coal to biofuels and beyond, *Atmos. Environ.*, vol. 43, no. 1, pp. 23–36, 2009.
- [2] B. Seshadri, N. S. Bolan, R. Naidu, H. Wang, and K. Sajwan, Clean coal technology combustion products: Properties, agricultural and environmental applications, and risk management, in *Advances in Agronomy*, D. L. Sparks, Ed. Amsterdam, The Netherlands: Elsevier, 2013, pp. 309–370.
- [3] G. A. Ryabov, O. M. Folomeev, D. S. Litun, D. A. Sankin, and I. G. Dmitryukova, Prospects for using the technology of circulating fluidized bed for technically refitting Russian thermal power stations, *Therm. Eng.*, vol. 56, no. 1, pp. 31–40, 2009.
- [4] J. Liang, H. Guo, K. Yu, B. Qu, C. Yue, and K. Qiao, An improved composite differential evolutionary algorithm with self-adaptive mutation strategy for identifying photovoltaic model parameters, in *Proc. 2021 5th Asian Conf. Artificial Intelligence Technology* (ACAIT), Haikou, China, 2022, pp. 591–599.
- [5] B. E. Launder and D. B. Spalding, The numerical computation of turbulent flows, *Comput. Meth. Appl. Mech. Eng.*, pp. 269–289, 1990.
- [6] X. Liu, G. Zhu, T. Asim, Y. Zhang, and R. Mishra, The innovative design of air caps for improving the thermal efficiency of CFB boilers, *Energy*, vol. 221, p. 119844, 2021.
- [7] S. Espatolero, L. M. Romeo, A. I. Escudero, and R. Kuivalainen, An operational approach for the designing of an energy integrated oxy-fuel CFB power plant, *Int. J. Greenh. Gas Contr.*, vol. 64, pp. 204–211, 2017.
- [8] G. Qi, X. Liu, Z. Wang, S. Zhang, J. Guan, H. Liu, and Z. Qu, Numerical simulation and optimization of heat-insulation material and structure for CFB boiler, *Therm. Sci. Eng. Prog.*, vol. 20, p. 100692, 2020.
- [9] J. Chen, W. Yin, S. Wang, G. Yu, J. Li, T. Hu, and F. Lin, Modelling of coal/biomass co-gasification in internal circulating fluidized bed using kinetic theory of granular mixture, *Energy Convers. Manag.*, vol. 148, pp. 506–516, 2017.
- [10] N. Qin and R. J. Li, Online simplified model and experimental comparison of CFB boiler thermal efficiency, *Appl. Therm. Eng.*, vol. 171, p. 115021, 2020.
- [11] M. L. Kubik, P. J. Coker, and J. F. Barlow, Increasing thermal plant flexibility in a high renewables power system, *Appl. Energy*, vol. 154, pp. 102–111, 2015.
- [12] Y. Gu, J. Xu, D. Chen, Z. Wang, and Q. Li, Overall review of peak shaving for coal-fired power units in China, *Renew. Sustain. Energy Rev.*, vol. 54, pp. 723–731, 2016.
- [13] F. Zhang, Y. Xue, D. Li, Z. Wu, and T. He, On the flexible operation of supercritical circulating fluidized bed: Burning carbon based decentralized active disturbance rejection control, *Energies*, vol. 12, no. 6, p. 1132, 2019.
- [14] J. Krzywanski and W. Nowak, Artificial intelligence treatment of SO₂ emissions from CFBC in air and oxygen-enriched conditions, *J. Energy Eng.*, vol. 142, no. 1, pp. 04015017-1–04015017-10, 2016.
- [15] Q. Wang, Z. Chen, J. Wang, L. Zeng, X. Zhang, X. Li, and Z. Li, Effects of secondary air distribution in primary combustion zone on combustion and NO_x emissions of a large-scale down-fired boiler with air staging, *Energy*, vol. 165, pp. 399–410, 2018.
- [16] G. J. Hesselmann, Optimization of combustion by fuel testing in a NO_x reduction test facility, *Fuel*, vol. 76, no. 13, pp. 1269–1275, 1997.
- [17] Ł. Śladewski, K. Wojdan, K. Świrski, T. Janda, D. Nabagło, and J. Chachuła, Optimization of combustion process in coal-fired power

plant with utilization of acoustic system for in-furnace temperature measurement, *Appl. Therm. Eng.*, vol. 123, pp. 711–720, 2017.

- [18] P. Madejski, Numerical study of a large-scale pulverized coal-fired boiler operation using CFD modeling based on the probability density function method, *Appl. Therm. Eng.*, vol. 145, pp. 352–363, 2018.
- [19] H. Gu, H. Zhu, Y. Cui, F. Si, R. Xue, H. Xi, and J. Zhang, Optimized scheme in coal-fired boiler combustion based on information entropy and modified K-prototypes algorithm, *Results Phys.*, vol. 9, pp. 1262–1274, 2018.
- [20] J. Pollak, A. Wozniak, Z. Dynia, and T. Lipanowicz, Effectiveness of artificial intelligence methods in applications to burning optimization and coal mills diagnostics on the basis of Iase's experiences in Turków power plant, (in Polish), *Inż. Chem. i Proces.*, vol. 25, no. 4, pp. 2265–2274, 2004.
- [21] F. Ren, Z. Li, Y. Zhang, S. Sun, X. Zhang, and Z. Chen, Influence of the secondary air-box damper opening on airflow and combustion characteristics of a down-fired 300-MW_e utility boiler, *Energy Fuels*, vol. 21, no. 2, pp. 668–676, 2007.
- [22] F. Fang, S. Yu, L. Wei, Y. Liu, and J. Liu, Data-driven control for combustion process of circulating fluidised bed boiler, *IET Cyberphys. Syst.*, vol. 5, no. 1, pp. 39–48, 2020.
- [23] S. Ouyang, J. Li, L. Dong, X. Zhou, and D. Yang, Calculation and analysis on the flow instability of an ultra supercritical boiler under low load, (in Chinese), *J. Xi'an Jiaotong Univ.*, vol. 53, no. 7, pp. 84–91, 2019.
- [24] J. Smrekar, M. Assadi, M. Fast, I. Kuštrin, and S. De, Development of artificial neural network model for a coal-fired boiler using real plant data, *Energy*, vol. 34, no. 2, pp. 144–152, 2009.
- [25] X. Wang, Z. Zhu, K. Wang, K. Yu, and Q. Lyu, Experimental study of pilot-scale CFB gasification: Effect of gasifying agent and coal feeding modes on the gasification performance, *Fuel*, vol. 251, pp. 603–610, 2019.
- [26] S. Zhu, M. Zhang, Y. Huang, Y. Wu, H. Yang, J. Lyu, X. Gao, F. Wang, and G. Yue, Thermodynamic analysis of a 660 MW ultrasupercritical CFB boiler unit, *Energy*, vol. 173, pp. 352–363, 2019.
- [27] Y. Wang, X. Qiu, Z. Zhou, Y. Duan, L. Li, J. Dai, H. Lin, Y. Luo, Z. Sun, and L. Duan, Ash deposition mechanism of shoe manufacturing waste combustion in a full-scale CFB boiler, *Fuel Process. Technol.*, vol. 221, p. 106948, 2021.
- [28] D. Shang, B. Li, Y. Li, and Z. Song, Effect of combustion optimization adjustment of 1000 MW ultra supercritical boilers on NO_x emission and boiler efficiency, (in Chinese), *J. Eng. Therm. Energy Power*, vol. 32, no. 3, pp. 61–68, 2017.
- [29] Y. Levron and D. Shmilovitz, Power systems' optimal peakshaving applying secondary storage, *Electr. Power Syst. Res.*, vol. 89, pp. 80–84, 2012.
- [30] M. Uddin, M. F. Romlie, M. F. Abdullah, S. Abd Halim, A. H. Abu Bakar, and T. Chia Kwang, A review on peak load shaving strategies, *Renew. Sustain. Energy Rev.*, vol. 82, pp. 3323–3332, 2018.
- [31] K. Qiao, J. Liang, K. Yu, M. Wang, B. Qu, C. Yue, and Y. Guo, A self-adaptive evolutionary multi-task based constrained multiobjective evolutionary algorithm, *IEEE Trans. Emerg. Top. Comput. Intell.*, doi: 10.1109/TETCI.2023.3236633.
- [32] K. Qiao, K. Yu, B. Qu, J. Liang, H. Song, C. Yue, H. Lin, and K. C. Tan, Dynamic auxiliary task-based evolutionary multitasking for constrained multi-objective optimization, *IEEE Trans. Evol. Comput.*, doi: 10.1109/TEVC.2022.3175065.
- [33] K. Qiao, K. Yu, B. Qu, J. Liang, H. Song, and C. Yue, An evolutionary multitasking optimization framework for constrained multiobjective optimization problems, *IEEE Trans. Evol. Comput.*, vol. 26, no. 2, pp. 263–277, 2022.
- [34] Z. Zhen, Study of the circulating fluidized bed boiler combustion optimization based on the improved wind driven optimization algorithm, PhD Thesis, Yanshan University, Qinhuangdao, China, 2017.
- [35] Y. Qiao, Z. Gao, and Z. Dong, Boiler combustion control system of

thermal power plant based on intelligent algorithm and system simulation analysis, (in Chinese), *Sci. Technol. Innov.*, vol. 2022, no. 35, pp. 185–188, 2022.

- [36] B. Chen, G. Cao, Y. Huang, J. Yue, W. Xu, Y. Wang, Y. Li, and B. Jin, Online calculation of coal-fired boiler combustion efficiency based on machine learning, (in Chinese), *Clean Coal Technol.*, vol. 27, no. 4, pp. 174–179, 2021.
- [37] Y. Tang, Q. Li, and H. Chen, Research and application of boiler NO_x emission based on BP neural network optimized by genetic algorithm, (in Chinese), *Shanxi Electr. Power*, vol. 2021, no. 2, pp. 56–59, 2021.
- [38] S. Elahifar, E. Assareh, and R. Moltames, Exergy Analysis and Thermodynamic Optimization of a steam power plant-Based Rankine cycle system Using Intelligent Optimization Algorithms, *Aust. J. Mech. Eng.*, vol. 20, no. 1, pp. 54–65, 2022.
- [39] P. Tan, B. He, C. Zhang, D. Rao, S. Li, Q. Fang, and G. Chen, Dynamic modeling of NO_x emission in a 660 MW coal-fired boiler with long short-term memory, *Energy*, vol. 176, pp. 429–436, 2019.
- [40] R. Dimeo and K. Y. Lee, Boiler-turbine control system design using a genetic algorithm, *IEEE Trans. Energy Convers.*, vol. 10, no. 4, pp. 752–759, 1995.
- [41] A. Chaibakhsh, A. Ghaffari, and S. Ali A Moosavian, A simulated model for a once-through boiler by parameter adjustment based on genetic algorithms, *Simul. Model. Pract. Theory*, vol. 15, no. 9, pp. 1029–1051, 2007.
- [42] S. Dal Secco, O. Juan, M. Louis-Louisy, J. Y. Lucas, P. Plion, and L. Porcheron, Using a genetic algorithm and CFD to identify low NO_x configurations in an industrial boiler, *Fuel*, vol. 158, pp. 672–683, 2015.
- [43] M. Zhu, J. Zhou, S. Su, J. Xu, A. Li, L. Chen, Y. Wang, S. Hu, L. Jiang, and J. Xiang, Study on supercritical CO₂ coal-fired boiler based on improved genetic algorithm, *Energy Convers. Manag.*, vol. 221, p. 113163, 2020.
- [44] G. N. Kumar and E. Gundabattini, Investigation of supercritical power plant boiler combustion process optimization through CFD and genetic algorithm methods, *Energies*, vol. 15, no. 23, p. 9076, 2022.
- [45] H. Pan, W. Zhong, Z. Wang, and G. Wang, Optimization of industrial boiler combustion control system based on genetic algorithm, *Comput. Electr. Eng.*, vol. 70, pp. 987–997, 2018.
- [46] E. Rosado-Tamariz, M. A. Zuniga-Garcia, and R. Batres, Optimization of a drum boiler startup using dynamic simulation and a micro-genetic algorithm, *Energy Rep.*, vol. 6, pp. 410–416, 2020.
- [47] S. Guo and J. Liang, Research on boiler temperature field reconstruction algorithm based on genetic algorithm, in *Proc. 2017 Int. Conf. Computer Technology, Electronics and Communication* (*ICCTEC*), Dalian, China, 2019, pp. 682–685.
- [48] P. Ilamathi, V. Selladurai, and K. Balamurugan, Modeling and optimization of unburned carbon in coal-fired boiler using artificial neural network and genetic algorithm, *J. Energy Resour. Technol.*, vol. 135, no. 3, p. 032201, 2013.
- [49] G. I. Tsoumalis, Z. N. Bampos, G. V. Chatzis, P. N. Biskas, and S. D. Keranidis, Minimization of natural gas consumption of domestic boilers with convolutional, long-short term memory neural networks and genetic algorithm, *Appl. Energy*, vol. 299, p. 117256, 2021.
- [50] X. Peng and P. Wang, An improved multiobjective genetic algorithm in optimization and its application to high efficiency and low NO_x emissions combustion, in *Proc. 2009 Asia-Pacific Power* and Energy Engineering Conf., Wuhan, China, 2009, pp. 1–4.
- [51] F. Wu, H. Zhou, T. Ren, L. Zheng, and K. Cen, Combining support vector regression and cellular genetic algorithm for multi-objective optimization of coal-fired utility boilers, *Fuel*, vol. 88, no. 10, pp. 1864–1870, 2009.
- [52] Y. Hu, C. K. Tan, J. Broughton, P. A. Roach, and L. Varga, Modelbased multi-objective optimisation of reheating furnace operations

using genetic algorithm, *Energy Procedia*, vol. 142, pp. 2143–2151, 2017.

- [53] R. Jha, P. K. Sen, and N. Chakraborti, Multi-objective genetic algorithms and genetic programming models for minimizing input carbon rates in a blast furnace compared with a conventional analytic approach, *Steel Res. Int.*, vol. 85, no. 2, pp. 219–232, 2014.
- [54] F. Wu, H. Zhou, J. P. Zhao, and K. F. Cen, A comparative study of the multi-objective optimization algorithms for coal-fired boilers, *Expert Syst. Appl.*, vol. 38, no. 6, pp. 7179–7185, 2011.
- [55] J. Krzywanski, H. Fan, Y. Feng, A. R. Shaikh, M. Fang, and Q. Wang, Genetic algorithms and neural networks in optimization of sorbent enhanced H₂ production in FB and CFB gasifiers, *Energy Convers. Manag.*, vol. 171, pp. 1651–1661, 2018.
- [56] M. V. J. J. Suresh, K. S. Reddy, and A. K. Kolar, ANN-GA based optimization of a high ash coal-fired supercritical power plant, *Appl. Energy*, vol. 88, no. 12, pp. 4867–4873, 2011.
- [57] Z. Wang, J. Li, and W. Sun, Boiler combustion optimization based on neural network and genetic algorithm, (in Chinese), J. North China Electr. Power Univ., vol. vol35, no. 1, pp. 14–17, 2008.
- [58] H, Zhou, K. Cen, and J. Fan, Modeling and optimization of the NO_x emission characteristics of a tangentially fired boiler with artificial neural networks, *Energy*, vol. 29, no. 1, pp. 167–183, 2004.
- [59] P. Chaudhari and S. Garg, Multi-objective optimization of maleic anhydride circulating fluidized bed (CFB) reactors, *Chem. Eng. Res. Des.*, vol. 141, pp. 115–132, 2019.
- [60] D. Buche, P. Stoll, R. Dornberger, and P. Koumoutsakos, Multiobjective evolutionary algorithm for the optimization of noisy combustion processes, *IEEE Trans. Syst. Man Cybern. C Appl. Rev.*, vol. 32, no. 4, pp. 460–473, 2002.
- [61] C. Xu, J. Lv, and Y. Zheng, An experiment and analysis for a boiler combustion optimization on efficiency and NO_x emissions, (in Chinese), *Boiler Technol.*, vol. 37, no. 5, pp. 69–74, 2006.
- [62] Y. Shi, W. Zhong, X. Chen, A. B. Yu, and J. Li, Combustion optimization of ultra supercritical boiler based on artificial intelligence, *Energy*, vol. 170, pp. 804–817, 2019.
- [63] H. Zhou, L. Zheng, and K. Cen, Computational intelligence approach for NO_x emissions minimization in a coal-fired utility boiler, *Energy Convers. Manag.*, vol. 51, no. 3, pp. 580–586, 2010.
- [64] S. M. Jiao and P. Han, Chaotic PSO and its application in modeling of circulating fluidized bed boiler combustion system, in *Proc.* 2011 IEEE Int. Conf. Intelligent Computing and Integrated Systems (ICISS), Guilin, China, 2011, pp. 1–5.
- [65] H. Zhao, P. H. Wang, X. Peng, J. Qian, and Q. Wang, Constrained optimization of combustion at a coal-fired utility boiler using hybrid particle swarm optimization with invasive weed, in *Proc.* 2009 Int. Conf. Energy and Environment Technology, Guilin, China, 2009, pp. 564–567.
- [66] Y. Zhang, H. Zhang, and W. Zhang, Optimization of coal-fired boiler on LS-SVM model and PSO algorithms, in Proc. 2015 6th Int. Conf. Manufacturing Science and Engineering, Qingdao, China, pp. 329–334.
- [67] L. Y. Ma, Y. P. Ge, and X. Cao, Superheated steam temperature control based on improved recurrent neural network and simplified PSO algorithm, *Appl. Mech. Mater.*, vol. 128-129, pp. 1065–1069, 2011.
- [68] S. Dong, M. Liu, G. Li, Q. Duan, and Q. Cui, Modeling of boiler efficiency and NO_x emission based on asymmetric PSO-BP, in *Proc. 2020 IEEE Int. Conf. Information Technology, Big Data and Artificial Intelligence (ICIBA)*, Chongqing, China, 2020, pp. 1237–1241.
- [69] J. F. Tuttle, R. Vesel, S. Alagarsamy, L. D. Blackburn, and K. Powell, Sustainable NO_x emission reduction at a coal-fired power station through the use of online neural network modeling and particle swarm optimization, *Contr. Eng. Pract.*, vol. 93, p. 104167, 2019.
- [70] Q. Li, Q. He, and Z. Liu, Low NO_x combustion optimization based on partial dimension opposition-based learning particle swarm optimization, *Fuel*, vol. 310, p. 122352, 2022.

- [71] Z. Tang, X. Wu, S. Cao, and M. Yang, Modeling of the boiler NO_x emission with a data driven algorithm, *J. Chem. Eng. Japan*, vol. 51, no. 8, pp. 695–703, 2018.
- [72] J. Liang, H. Guo, K. Chen, K. Yu, C. Yue, and X. Li, An improved Kalman particle swarm optimization for modeling and optimizing of boiler combustion characteristics, *Robotica*, vol. 41, no. 4, pp. 1087–1097, 2023.
- [73] J. Yuan, X. Ren, Y. Xie, and Z. Li, Research on optimization of boiler air distribution system based on deep neural network, *J. Phys.: Conf. Ser.*, vol. 1624, no. 5, p. 052019, 2020.
- [74] Y. Lv, F. Hong, T. Yang, F. Fang, and J. Liu, A dynamic model for the bed temperature prediction of circulating fluidized bed boilers based on least squares support vector machine with real operational data, *Energy*, vol. 124, pp. 284–294, 2017.
- [75] H. Wang, G. Zhang, Y. Huang, and Y. Zhang, Study on boiler's comprehensive benefits optimization based on PSO optimized XGBoost algorithm, *E3S Web Conf.*, vol. 261, p. 01027, 2021.
- [76] X. Wu, Z. Tang, and S. Cao, A hybrid least square support vector machine for boiler efficiency prediction, in *Proc. 2017 IEEE 3rd Information Technology and Mechatronics Engineering Conference (ITOEC)*, Chongqing, China, 2017, pp. 1202–1205.
- [77] J. Luo, S. Chen, L. Wu, and S. Zhang, An optimal sparseness approach for least square support vector machine, in *Proc. 26th Chinese Control and Decision Conference (2014 CCDC)*, Changsha, China, 2014, pp. 3621–3626.
- [78] W. Fan, F. Si, S. Ren, C. Yu, Y. Cui, and P. Wang, Integration of continuous restricted Boltzmann machine and SVR in NO_x emissions prediction of a tangential firing boiler, *Chemom. Intell. Lab. Syst.*, vol. 195, p. 103870, 2019.
- [79] H. Zhao, P. H. Wang, J. Qian, and X. Y. Peng, An improved particle swarm algorithm and its application in low NO_x combustion optimization of coal-fired utility boiler, in *Proc. 2010 Asia-Pacific Power and Energy Engineering Conf.*, Chengdu, China, 2010, pp. 1–4.
- [80] A. Daraz, S. A. Malik, H. Mokhlis, I. U. Haq, G. F. Laghari, and N. N. Mansor, Fitness dependent optimizer-based automatic generation control of multi-source interconnected power system with non-linearities, *IEEE Access*, vol. 8, pp. 100989–101003, 2020.
- [81] H. Zhao and P. H. Wang, Modeling and optimization of efficiency and NO_x emission at a coal-fired utility boiler, in *Proc. 2009 Asia-Pacific Power and Energy Engineering Conf.*, Wuhan, China, 2009, pp. 1–4.
- [82] W. Xu, Y. Huang, S. Song, Y. Chen, G. Cao, M. Yu, B. Chen, R. Zhang, Y. Liu, and Y. Zou, A new online optimization method for boiler combustion system based on the data-driven technique and the case-based reasoning principle, *Energy*, vol. 263, p. 125508, 2023.
- [83] A. B. V. Barboza, S. Mohan, and P. Dinesha, On reducing the emissions of CO, HC, and NO_x from gasoline blended with hydrogen peroxide and ethanol: Optimization study aided with ANN-PSO, *Environ. Pollut.*, vol. 310, p. 119866, 2022.
- [84] T. Ye, M. Dong, J. Long, Y. Zheng, Y. Liang, and J. Lu, Multiobjective modeling of boiler combustion based on feature fusion and Bayesian optimization, *Comput. Chem. Eng.*, vol. 165, p. 107913, 2022.
- [85] W. Xu, Y. Huang, S. Song, B. Chen, and X. Qi, A novel online combustion optimization method for boiler combining dynamic modeling, multi-objective optimization and improved case-based reasoning, *Fuel*, vol. 337, p. 126854, 2023.
- [86] P. Garg and M. S. Orosz, Economic optimization of Organic Rankine cycle with pure fluids and mixtures for waste heat and solar applications using particle swarm optimization method, *Energy Convers. Manag.*, vol. 165, pp. 649–668, 2018.
- [87] H. Jagtap, A. Bewoor, R. Kumar, M. H. Ahmadi, and G. Lorenzini, Markov-based performance evaluation and availability optimization of the boiler-furnace system in coal-fired thermal power plant using PSO, *Energy Rep.*, vol. 6, pp. 1124–1134, 2020.

- [88] M. Sayed, S. M. Gharghory, and H. A. Kamal, Gain tuning PI controllers for boiler turbine unit using a new hybrid jump PSO, *J. Electr. Syst. Inf. Technol.*, vol. 2, no. 1, pp. 99–110, 2015.
- [89] X. Xu, Q. Chen, M. Ren, L. Cheng, and J. Xie, Combustion optimization for coal fired power plant boilers based on improved distributed ELM and distributed PSO, *Energies*, vol. 12, no. 6, p. 1036, 2019.
- [90] H. Aygun and H. Demirel, Comparison of PSO-PID, FLC and PID in a circulating fluidized bed boiler, in *Proc. 2011 7th Int. Conf. Electrical and Electronics Engineering (ELECO)*, Bursa, Turkey, 2011, pp. II-376–II-380.
- [91] L. Zheng, M. Yu, and S. Yu, Support vector regression and ant colony optimization for combustion performance of boilers, in *Proc. 2008 Fourth Int. Conf. Natural Computation*, Jinan, China, 2008, pp. 178–182.
- [92] Z. Tang, S. Wang, X. Chai, S. Cao, T. Ouyang, and Y. Li, Autoencoder-extreme learning machine model for boiler NO_x emission concentration prediction, *Energy*, vol. 256, p. 124552, 2022.
- [93] Q. Li, J. Wu, and H. Wei, Reduction of elemental mercury in coalfired boiler flue gas with computational intelligence approach, *Energy*, vol. 160, pp. 753–762, 2018.
- [94] W. Long, X. Liang, Z. Long, and Z. Li, Predictive control based on LSSVM and ACO for boiler combustion optimization, (in Chinese), *Electr. Power Autom. Equip.*, vol. 31, no. 11, pp. 89–93, 2011.
- [95] Q. Ruan, W. Pan, S. Yan, and T. Wu, Control of CFB boiler combustion system by improving ant colony PID—neural decoupling, (in Chinese), *Comput. Meas. Control*, vol. 23, no. 12, pp. 4023–4025, 2015.
- [96] B. Xiao, R. Chen, C. Lu, and Y. Li, Superheated steam temperature control based-on cultural based ant colony optimization algorithm for power station boiler, in *Proc. 32nd Chinese Control Conference*, Xi'an, China, 2013, pp. 7965–7970.
- [97] Q. Xu, Y. Wang, W. Lin, H. Liang, and J. Wan, Calculation of CO₂ emissions in boilers based on ACO-LSSVM method for carbon element analysis of coal, (in Chinese), *J. Fuzhou Univ. (Nat. Sci. Ed.*), vol. 43, no. 4, pp. 548–553, 2015.
- [98] Z. Zhang, J. Qian, and X. Feng, Prediction method of unburned carbon content in fly ash based on ant colony algorithm, (in Chinese), *Meas. Technol.*, vol. 37, no. 1, pp. 18–20, 2017.
- [99] K. Jagatheesan, B. Anand, K. Baskaran, N. Dey, A. S. Ashour, and V. E. Balas, Effect of nonlinearity and boiler dynamics in automatic generation control of multi-area thermal power system with proportional-integral-derivative and ant colony optimization technique, in *Recent Advances in Nonlinear Dynamics and Synchronization*, K. Kyamakya, W. Mathis, R. Stoop, J. C. Chedjou, and Z. Li Eds. Cham, Germany: Springer, 2018: pp. 89–110.
- [100] Q. Xu, J. Yang, and Y. Yang, Identification and control of boiler combustion system based on neural networks and ant colony optimization algorithm, in *Proc. 2008 7th World Congress on Intelligent Control and Automation*, Chongqing, 2008, pp. 765–768.
- [101] R. Chen, H. H. Yue, R. Yue, Y. Ai, and J. X. Zheng, Numerical simulation of combustion in a biomass briquette chain boiler, *Biomass Convers. Biorefin.*, vol. 11, no. 5, pp. 1521–1536, 2021.
- [102] J. Kaliannan, A. Baskaran, N. Dey, and A. S. Ashour, Ant colony optimization algorithm based PID controller for LFC of single area power system with non-linearity and boiler dynamics, *World J. Model. Simul.*, vol. 12, no. 1, pp. 3–14, 2016.
- [103] F. Wu, H. Zhou, L. Zheng, and K. Cen, Application of scaleable chaotic ant colony algorithm in control of unburned carbon in fly ash, (in Chinese), *J. Zhejiang Univ. Eng. Sci.*, vol. 44, no. 6, pp. 1127–1132, 2010.
- [104] L. D. Blackburn, J. F. Tuttle, K. Andersson, J. D. Hedengren, and K. M. Powell, Dynamic machine learning-based optimization algorithm to improve boiler efficiency, *J. Process. Contr.*, vol. 120, pp. 129–149, 2022.

- [105] M. Lan, Y. Li, G. Zhao, Y. Zhou, Z. Jiang, and Y. Gan, Study on boiler combustion modeling based on MAPSO optimizing LSSVM model parameters, (in Chinese), *J. Central South Univ. Sci. Technol.*, vol. 53, no. 4, pp. 1506–1515, 2022.
- [106] L. G. Zheng, H. Zhou, K. F. Cen, and C. L. Wang, A comparative study of optimization algorithms for low NO_x combustion modification at a coal-fired utility boiler, *Expert Syst. Appl.*, vol. 36, no. 2, pp. 2780–2793, 2009.
- [107] H. Zhou, J. Zhao, L. Zheng, C. Wang, and K. Cen, Modeling NO_x emissions from coal-fired utility boilers using support vector regression with ant colony optimization, *Eng. Appl. Artif. Intell.*, vol. 25, no. 1, pp. 147–158, Feb, 2012.
- [108] L. Zheng, H. Zhou, C. Wang, and K. Cen, Combining support vector regression and ant colony optimization to reduce NO_x emissions in coal-fired utility boilers, *Energy Fuels*, vol. 22, no. 2, pp. 1034–1040, 2008.
- [109] R. V. Rao, Teaching-learning-based optimization algorithm, in *Teaching Learning Based Optimization Algorithm*, R. V. Rao Ed. Cham, Germany: Springer, 2016: pp. 9–39.
- [110] A. Shabanpour-Haghighi and A. R. Seifi, Simultaneous integrated optimal energy flow of electricity, gas, and heat, *Energy Convers. Manag.*, vol. 101, pp. 579–591, 2015.
- [111] G. Li, P. Niu, W. Zhang, and Y. Liu, Model NO_x emissions by least squares support vector machine with tuning based on ameliorated teaching-learning-based optimization, *Chemom. Intell. Lab. Syst.*, vol. 126, pp. 11–20, 2013.
- [112] P. Niu, Y. Ma, M. Li, S. Yan, and G. Li, A kind of parameters selfadjusting extreme learning machine, *Neural Process. Lett.*, vol. 44, no. 3, pp. 813–830, 2016.
- [113] R. Azizipanah-Abarghooee, T. Niknam, F. Bavafa, and M. Zare, Short-term scheduling of thermal power systems using hybrid gradient based modified teaching-learning optimizer with black hole algorithm, *Electr. Power Syst. Res.*, vol. 108, pp. 16–34, 2014.
- [114] A. Khosravi, M. Malekan, J. J. G. Pabon, X. Zhao, and M. E. H. Assad, Design parameter modelling of solar power tower system using adaptive neuro-fuzzy inference system optimized with a combination of genetic algorithm and teaching learning-based optimization algorithm, *J. Clean. Prod.*, vol. 244, p. 118904, 2020.
- [115] M. Basu, Teaching-learning-based optimization algorithm for multiarea economic dispatch, *Energy*, vol. 68, pp. 21–28, 2014.
- [116] Y. Ma, P. Niu, K. Chen, S. Yan, and G. Li, Optimize NO_x emissions model of boiler based on chaos group teaching-learning based optimization algorithm, (in Chinese), *Acta Metrol. Sin.*, vol. 39, no. 1, pp. 125–129, 2018.
- [117] R. K. Sahu, T. S. Gorripotu, and S. Panda, Automatic generation control of multi-area power systems with diverse energy sources using Teaching Learning Based Optimization algorithm, *Eng. Sci. Technol. Int. J.*, vol. 19, no. 1, pp. 113–134, 2016.
- [118] Y. Ma, H. Tang, H. Wang, Z. Wang, X. Zhang, and L. Li, A novel chaotic teaching learning based optimization algorithm and its application in optimization of extreme learning machine, *J. Phys.:Conf. Ser.*, vol. 2003, no. 1, p. 012003, 2021.
- [119] D. Keihan Asl, A. R. Seifi, M. Rastegar, and M. Mohammadi, Multi-objective optimal operation of integrated thermal-natural gaselectrical energy distribution systems, *Appl. Therm. Eng.*, vol. 181, p. 115951, 2020.
- [120] Y. Ma, H. Wang, X. Zhang, L. Hou, and J. Song, Three-objective optimization of boiler combustion process based on multi-objective teaching–learning based optimization algorithm and ameliorated extreme learning machine, *Mach. Learn. Appl.*, vol. 5, p. 100082, 2021.
- [121] S. Liu, Y. Ma, R. Wang, W. Dong, and Y. Wang, Optimize the NO_x emission concentration of Circulation Fluidized Bed Boiler based on on-line learning neural network and modified TLBO algorithm, in *Proc. 2022 14th Int. Conf. Machine Learning and Computing* (*ICMLC*), Guangzhou China, pp. 83–90.
- [122] F. Ahmed, J. K. Kim, A. U. Khan, H. Y. Park, and Y. K. Yeo, A fast converging and consistent teaching-learning-self-study

CAAI Artificial Intelligence Research

algorithm for optimization: A case study of tuning of LSSVM parameters for the prediction of NO_x emissions from a tangentially fired pulverized coal boiler, *J. Chem. Eng. Japan*, vol. 50, no. 4, pp. 273–290, 2017.

- [123] X. Li, P. Niu, G. Li, and J. Liu, An adaptive extreme learning machine for modeling NO_x emission of a 300 MW circulating fluidized bed boiler, *Neural Process. Lett.*, vol. 46, no. 2, pp. 643–662, 2017.
- [124] Y. Ma, X. Zhang, J. Song, and L. Chen, A modified teaching-learning-based optimization algorithm for solving optimization problem, *Knowl. Based Syst.*, vol. 212, p. 106599, 2021.
- [125] Z. Tang, Y. Li, X. Chai, H. Zhang, and S. Cao, Adaptive nonlinear model predictive control of NO_x emissions under load constraints in power plant boilers, *J. Chem. Eng. Japan*, vol. 53, no. 1, pp. 36–44, 2020.
- [126] H. Zhang, C. Gao, and P. Ren, Prediction model of boiler NO_x emission based on least squares support vector machine, (in Chinese), *Jilin Electr. Power*, vol. 47, no. 3, pp. 18–20, 2019.
- [127] Q. Li, H. Zhang, D. Peng, Y. Guo, N. Wang, and Y. Sun, Multivariable modeling research for main-steam temperature of power station boiler based on improved differential evolution algorithm, (in Chinese), J. Syst. Simul., vol. 29, no. 8, pp. 1712–1718, 2017.
- [128] Z. Tang and H. Zhang, Modeling NO_x emission of coal-fired boiler with differential evolution optimized least square support vector machine, in *Proc. 2018 Chinese Control and Decision Conference* (*CCDC*), Shenyang, China, 2018, pp. 3364–3367.
- [129] Z. Shen and Q. Li, Model NO_x emissions emitted from coal-fired boilers based on differential evolution fast learning network, *J. Phys.:Conf. Ser.*, vol. 1624, no. 2, p. 022029, 2020.
- [130] H. Li, Y. Zheng, Y. Xia, and F. Liu, Optimization of SCR denitration spraying based on support vector machine and differential evolution algorithm, *IOP Conf. Ser.:Earth Environ. Sci.*, vol. 218, p. 012130, 2019.
- [131] J. Song, C. E. Romero, Z. Yao, and B. He, Improved artificial bee colony-based optimization of boiler combustion considering NO_x emissions, heat rate and fly ash recycling for on-line applications, *Fuel*, vol. 172, pp. 20–28, 2016.
- [132] X. H. Gao and Z. G. Su, Artificial bee colony optimization of NO_x emission and reheat steam temperature in a 1000 MW boiler, *Math. Probl. Eng.*, vol. 2019, pp. 1–13, 2019.
- [133] G. Li, P. Niu, Y. Ma, H. Wang, and W. Zhang, Tuning extreme learning machine by an improved artificial bee colony to model and optimize the boiler efficiency, *Knowl. Based Syst.*, vol. 67, pp. 278–289, 2014.
- [134] J. T. Wu, Y. B. Zhang, G. S. Xu, Y. Lin, and X. G. Lv, Research on the optimization of boiler efficiency based on artificial bee colony algorithm, *Comput. Inf. Sci.*, vol. 7, no. 4, p. 30, 2014.
- [135] G. Hou, L. Gong, Z. Yang, and J. Zhang, Multi-objective economic model predictive control for gas turbine system based on quantum simultaneous whale optimization algorithm, *Energy Convers. Manag.*, vol. 207, p. 112498, 2020.
- [136] C. Zhen and H. Liu, Model for predicting NO_x emission from boilers based on MWOA-LSSVM integration, *J. Chem. Eng. Japan*, vol. 52, no. 8, pp. 702–709, 2019.
- [137] S. B. Savargave and A. M. Deshpande, Self-adaptive whale optimization for the design and modelling of boiler plant, *Control Cybern.*, vol. 47, no. 4, p. 329, 2018.
- [138] Y. Zhao, Q. Wu, H. Li, S. Ma, P. He, J. Zhao, and Y. Li, Optimization of thermal efficiency and unburned carbon in fly ash of coal-fired utility boiler via grey wolf optimizer algorithm, *IEEE Access*, vol. 7, pp. 114414–114425, 2019.
- [139] P. Niu, C. Shi, N. Liu, Y. Ma, Z. Wu, and J. Li, Prediction model for boiler NO_x emission based on adaptive quantum grey wolf optimization, (in Chinese), *J. Chinese Soc. Power Eng.*, vol. 38, no. 4, pp. 278–285, 2018.
- [140] Y. Qie, J. Wang, J. Liu, and Y. Liu, Prediction of export SO₂ concentration based on feature selection and MGWO-LSSVM, (in

Chinese), Energy Conserv., vol. 40, no. 9, pp. 33-37, 2021.

- [141] X. Li, P. Niu, and J. Liu, Combustion optimization of a boiler based on the chaos and Lévy flight vortex search algorithm, *Appl. Math. Model.*, vol. 58, pp. 3–18, 2018.
- [142] G. Li and B. Shi, Optimization for boiler based on data mining and multi-condition combustion model, in *Proc. 2021 China Automation Congress (CAC)*, Beijing, China, 2022, pp. 6976–6982.
- [143] P. Tan, J. Xia, C. Zhang, Q. Fang, and G. Chen, Modeling and optimization of NO_x emission in a coal-fired power plant using advanced machine learning methods, *Energy Procedia*, vol. 61, pp. 377–380, 2014.
- [144] Z. Tang and Z. Zhang, The multi-objective optimization of combustion system operations based on deep data-driven models, *Energy*, vol. 182, pp. 37–47, 2019.
- [145] Y. Ma, C. Xu, H. Wang, R. Wang, S. Liu, and X. Gu, Model NO_x, SO₂ emissions concentration and thermal efficiency of CFBB based on a hyper-parameter self-optimized broad learning system, *Energies*, vol. 15, no. 20, p. 7700, 2022.
- [146] L. Pattanayak, S. P. K. Ayyagari, and J. N. Sahu, Optimization of sootblowing frequency to improve boiler performance and reduce combustion pollution, *Clean Technol. Environ. Policy*, vol. 17, no. 7, pp. 1897–1906, 2015.
- [147] A. A. M. Rahat, C. Wang, R. M. Everson, and J. E. Fieldsend, Datadriven multi-objective optimisation of coal-fired boiler combustion systems, *Appl. Energy*, vol. 229, pp. 446–458, 2018.
- [148] Y. Gu, W. Zhao, and Z. Wu, An optimal MVs decision model for boiler combustion optimization, (in Chinese), *Proc. CSEE*, vol. 32, no. 2, pp. 39–44, 2012.
- [149] G. Mirmoshtaghi, J. Skvaril, P. E. Campana, H. Li, E. Thorin, and E. Dahlquist, The influence of different parameters on biomass gasification in circulating fluidized bed gasifiers, *Energy Convers. Manag.*, vol. 126, pp. 110–123, 2016.
- [150] A. Kusiak and Z. Song, Combustion efficiency optimization and virtual testing: a data-mining approach, *IEEE Trans. Ind. Inform.*, vol. 2, no. 3, pp. 176–184, 2006.
- [151] B. S. Kim, T. Y. Kim, T. C. Park, and Y. K. Yeo, Comparative study of estimation methods of NO_x emission with selection of input parameters for a coal-fired boiler, *Korean J. Chem. Eng.*, vol. 35, no. 9, pp. 1779–1790, 2018.
- [152] F. Wang, S. Ma, H. Wang, Y. Li, Z. Qin, and J. Zhang, A hybrid model integrating improved flower pollination algorithm-based feature selection and improved random forest for NO_x emission estimation of coal-fired power plants, *Measurement*, vol. 125, pp. 303–312, 2018.
- [153] S. A. Kalogirou, Applications of artificial neural networks in energy systems, *Energy Convers. Manag.*, vol. 40, no. 10, pp. 1073–1087, 1999.
- [154] M. Talaat, M. H. Gobran, and M. Wasfi, A hybrid model of an artificial neural network with thermodynamic model for system diagnosis of electrical power plant gas turbine, *Eng. Appl. Artif. Intell.*, vol. 68, pp. 222–235, 2018.
- [155] S. S. Gururajapathy, H. Mokhlis, and H. A. Illias, Fault location and detection techniques in power distribution systems with distributed generation: A review, *Renew. Sustain. Energy Rev.*, vol. 74, pp. 949–958, 2017.
- [156] P. Costamagna, A. De Giorgi, G. Moser, S. B. Serpico, and A. Trucco, Data-driven techniques for fault diagnosis in power generation plants based on solid oxide fuel cells, *Energy Convers. Manag.*, vol. 180, pp. 281–291, 2019.
- [157] W. Zhang, Y. Zhang, X. Bai, J. Liu, D. Zeng, and T. Qiu, A robust fuzzy tree method with outlier detection for combustion models and optimization, *Chemom. Intell. Lab. Syst.*, vol. 158, pp. 130–137,Nov, 2016.
- [158] Y. Jung, J. Jung, B. Kim, and S. Han, Long short-term memory recurrent neural network for modeling temporal patterns in longterm power forecasting for solar PV facilities: Case study of South Korea, J. Clean. Prod., vol. 250, p. 119476, 2020.
- [159] Z. F. Liu, L. L. Li, M. L. Tseng, and M. K. Lim, Prediction short-

term photovoltaic power using improved chicken swarm optimizer -Extreme learning machine model, *J. Clean. Prod.*, vol. 248, p. 119272, 2020.

- [160] G. Rubio, H. Pomares, I. Rojas, and L. J. Herrera, A heuristic method for parameter selection in LS-SVM: Application to time series prediction, *Int. J. Forecast.*, vol. 27, no. 3, pp. 725–739, 2011.
- [161] T. Palmé, M. Fast, and M. Thern, Gas turbine sensor validation through classification with artificial neural networks, *Appl. Energy*, vol. 88, no. 11, pp. 3898–3904, 2011.
- [162] X. L. Chen, P.-H. Wang, Y. S. Hao, and M. Zhao, Evidential KNNbased condition monitoring and early warning method with applications in power plant, *Neurocomputing*, vol. 315, pp. 18–32, 2018.
- [163] K. Karabacak and N. Cetin, Artificial neural networks for controlling wind-PV power systems: A review, *Renew. Sustain. Energy Rev.*, vol. 29, pp. 804–827, 2014.
- [164] Y. Zhang, T. Luan, and Y. Yao, MtsPSO-PID neural network decoupling control in power plant boiler, *IFAC Proc. Vol.*, vol. 46, no. 20, pp. 101–105, 2013.
- [165] L. Ma, Z. Wang, and K. Y. Lee, Neural network inverse control for the coordinated system of a 600MW supercritical boiler unit, *IFAC Proc. Vol.*, vol. 47, no. 3, pp. 999–1004, 2014.
- [166] L. Ma, K. Y. Lee, and Z. Wang, Intelligent coordinated controller design for a 600 MW supercritical boiler unit based on expandedstructure neural network inverse models, *Contr. Eng. Pract.*, vol. 53, pp. 194–201, 2016.
- [167] J. H. Sun, W. Wang, and K. W. Zheng, Research of PID neural networks decoupling control of marine nuclear power plant, (in Chinese), J. Harbin Eng. Univ., vol. 28, no. 6, pp. 656–659, 2007.
- [168] A. Mozaffari, M. Azimi, and M. Gorji-Bandpy, Ensemble mutable smart bee algorithm and a robust neural identifier for optimal design of a large scale power system, *J. Comput. Sci.*, vol. 5, no. 2, pp. 206–223, 2014.
- [169] H. Omrani, A. Alizadeh, and A. Emrouznejad, Finding the optimal combination of power plants alternatives: A multi response Taguchi-neural network using TOPSIS and fuzzy best-worst method, J. Clean. Prod., vol. 203, pp. 210–223, 2018.
- [170] B. Zhang, W. Hu, D. Cao, Q. Huang, Z. Chen, and F. Blaabjerg, Deep reinforcement learning-based approach for optimizing energy conversion in integrated electrical and heating system with renewable energy, *Energy Convers. Manag.*, vol. 202, p. 112199, 2019.
- [171] J. M. Molina, P. Isasi, A. Berlanga, and A. Sanchis, Hydroelectric power plant management relying on neural networks and expert system integration, *Eng. Appl. Artif. Intell.*, vol. 13, no. 3, pp. 357–369, 2000.
- [172] R. Cass and B. Radl, Adaptive process optimization using functional-link networks and evolutionary optimization, *IFAC Proc. Vol.*, vol. 29, no. 7, pp. 253–258, 1996.
- [173] M. Hossain, S. Mekhilef, M. Danesh, L. Olatomiwa, and S. Shamshirband, Application of extreme learning machine for short term output power forecasting of three grid-connected PV systems, *J. Clean. Prod.*, vol. 167, pp. 395–405, 2017.
- [174] P. Niu, Y. Ma, and G. Li, Model NO_x emission and thermal efficiency of CFBB based on an ameliorated extreme learning machine, *Soft Comput.*, vol. 22, no. 14, pp. 4685–4701, 2018.
- [175] G. Q. Li, X. B. Qi, K. C. C. Chan, and B. Chen, Deep bidirectional learning machine for predicting NO_x emissions and boiler efficiency from a coal-fired boiler, *Energy Fuels*, vol. 31, no. 10, pp. 11471–11480, 2017.
- [176] G. Yang, Y. Wang, and X. Li, Prediction of the NO_x emissions from thermal power plant using long-short term memory neural network, *Energy*, vol. 192, p. 116597, 2020.
- [177] W. Muhammad Ashraf, G. Moeen Uddin, S. Muhammad Arafat, S. Afghan, A. Hassan Kamal, M. Asim, M. Haider Khan, M. Waqas Rafique, U. Naumann, S. G. Niazi, et al., Optimization of a 660 MW_e supercritical power plant performance—A case of industry

4.0 in the data-driven operational management part 1. thermal efficiency, *Energies*, vol. 13, no. 21, p. 5592, 2020.

- [178] F. Kartal and U. Özveren, Investigation of an integrated circulating fluidized bed gasifier/steam turbine/proton exchange membrane (PEM) fuel cell system for torrefied biomass and modeling with artificial intelligence approach, *Energy Convers. Manag.*, vol. 263, p. 115718, 2022.
- [179] J. Grochowalski, P. Jachymek, M. Andrzejczyk, M. Klajny, A. Widuch, P. Morkisz, B. Hernik, J. Zdeb, and W. Adamczyk, Towards application of machine learning algorithms for prediction temperature distribution within CFB boiler based on specified operating conditions, *Energy*, vol. 237, p. 121538, 2021.
- [180] R. C. Booth and W. B. Roland, Neural network-based combustion optimization reduces NO_x emissions while improving performance, in *Proc. IEEE Industry Applications on Dynamic Modeling Control Applications for Industry Workshop*, Vancouver, Canada, 2002, pp. 1–6.
- [181] J. Wang, Z. Zhang, and S. Jiang, SmartProcess combustion optimization system and its application, (in Chinese), *Electr. Equip.*, vol. 7, no. 2, pp. 28–30, 2006.
- [182] Y. Pan, Research on the optimization technology of boiler combustion in power plant based on intelligence algorithm, PhD thesis, North China Electric Power University, Beijing, China, 2015.
- [183] H. Zhou, J. Mao, Z. Chi, X. Jiang, Z. Wang, and K. Cen, Predicting low NO_x combustion property of a coal-fired boiler, (in Chinese), *Environ. Sci.*, vol. 23, no. 2, pp. 18–22, 2002.
- [184] Y. Tunckaya and E. Koklukaya, Comparative prediction analysis of 600 MWe coal-fired power plant production rate using statistical and neural-based models, *J. Energy Inst.*, vol. 88, no. 1, pp. 11–18, 2015.
- [185] P. Niu, X. Xiao, G. Li, Y. Ma, G. Chen, and X. Zhang, Parameter optimization for NO_x emission model of power plant boilers based on gravitational search algorithm, (in Chinese), *J. Chinese Soc. Power Eng.*, vol. 33, no. 2, pp. 100–106, 2013.
- [186] P. Niu, H. Ma, G. Li, Y. Ma, G. Chen, and X. Zhang, Study on NO_x emission from cfb boilers based on support vector, (in Chinese), J. *Chinese Soc. Power Eng.*, vol. 33, no. 4, pp. 267–271, 2013.
- [187] X. Tong, W. Sun, and Q. Li, A modified PSO-LSSVM algorithm in boiler combustion optimization application, (in Chinese), *Manuf. Autom.*, vol. 37, no. 2, pp. 12–15, 2015.
- [188] Q. Song and Y. Hou, Modeling optimization for boiler based on modified fruit fly algorithm, (in Chinese), *Comput. Simul.*, vol. 35, no. 1, pp. 98–102, 2018.
- [189] Y. Lv, J. Liu, T. Yang, and D. Zeng, A novel least squares support vector machine ensemble model for NO_x emission prediction of a coal-fired boiler, *Energy*, vol. 55, pp. 319–329, 2013.
- [190] Y. Gu, W. Zhao, and Z. Wu, Online adaptive least squares support vector machine and its application in utility boiler combustion optimization systems, *J. Process. Contr.*, vol. 21, no. 7, pp. 1040–1048, 2011.
- [191] C. Wang, Y. Liu, S. Zheng, and A. Jiang, Optimizing combustion of coal fired boilers for reducing NO_x emission using Gaussian Process, *Energy*, vol. 153, pp. 149–158, 2018.
- [192] J. W. Chew and R. A. Cocco, Application of machine learning methods to understand and predict circulating fluidized bed riser flow characteristics, *Chem. Eng. Sci.*, vol. 217, p. 115503, 2020.
- [193] Z. Yuan, L. Meng, X. Gu, Y. Bai, H. Cui, and C. Jiang, Prediction of NO_x emissions for coal-fired power plants with stackedgeneralization ensemble method, *Fuel*, vol. 289, p. 119748, 2021.
- [194] Q. Wang, Application prospect analysis of NeuSIGHT neural network closed-loop combustion optimization control system in Shajiao B power plant, (in Chinese), *Henan Electr. Power*, vol. 33, no. 3, pp. 34–37, 2005.
- [195] L. Shen, Combustion optimization technology of coal-fired power plant boilers and related algorithm application research, PhD thesis, Zhejiang University, Hangzhou, China, 2011.
- [196] I. Lalak, J. Seeber, F. Kluger, and S. Krupka, Operational

CAAI Artificial Intelligence Research

experience with high efficiency cyclones: Comparison between boiler A and B in the Zeran power plant—Warsaw, Poland, in *Proc. 17th Int. Conf. Fluidized Bed Combustion*, Jacksonville, FL, USA, 2003, pp. 685–690.

- [197] J. Li, H. Yang, Y. Wu, J. Lv, and G. Yue, Effects of the updated national emission regulation in China on circulating fluidized bed boilers and the solutions to meet them, *Environ. Sci. Technol.*, vol. 47, no. 12, pp. 6681–6687, 2013.
- [198] R.-P. Nikula, M. Ruusunen, and K. Leiviskä, Data-driven framework for boiler performance monitoring, *Appl. Energy*, vol. 183, pp. 1374–1388, 2016.
- [199] C. Huang and X. Sheng, Data-driven model identification of boilerturbine coupled process in 1000 MW ultra-supercritical unit by improved bird swarm algorithm, *Energy*, vol. 205, p. 118009, 2020.
- [200] K. Qiao, K. Yu, B. Qu, J. Liang, C. Yue, and X. Ban, Feature extraction for recommendation of constrained multi-objective evolutionary algorithms, *IEEE Trans. Evol. Comput.*, doi: 10.1109/TEVC.2022.3186667.
- [201] Y. Cheng, Y. Huang, B. Pang, and W. Zhang, ThermalNet: A deep reinforcement learning-based combustion optimization system for coal-fired boiler, *Eng. Appl. Artif. Intell.*, vol. 74, pp. 303–311, 2018.
- [202] Z. Han, Y. Xie, M. M. Hossain, and C. Xu, A hybrid deep neural network model for NO_x emission prediction of heavy oil-fired boiler flames, *Fuel*, vol. 333, p. 126419, 2023.
- [203] F. Wang, S. Ma, H. Wang, Y. Li, and J. Zhang, Prediction of NO_x emission for coal-fired boilers based on deep belief network, *Contr. Eng. Pract.*, vol. 80, pp. 26–35, 2018.
- [204] Y. Bi, C. Chen, X. Huang, H. Wang, and G. Wei, Discrimination method of biomass slagging tendency based on particle swarm optimization deep neural network (DNN), *Energy*, vol. 262, p. 125368, 2023.
- [205] Z. Han, Y. Huang, J. Li, B. Zhang, M. M. Hossain, and C. Xu, A hybrid deep neural network based prediction of 300 MW coal-fired boiler combustion operation condition, *Sci. China Technol. Sci.*, vol. 64, no. 10, pp. 2300–2311, 2021.
- [206] Y. Lv, T. Yang, and J. Liu, An adaptive least squares support vector machine model with a novel update for NO_x emission prediction, *Chemometr. Intell. Lab. Syst.*, vol. 145, pp. 103–113, 2015.
- [207] V. Kovalnogov, R. Fedorov, V. Klyachkin, D. Generalov, Y.

Kuvayskova, and S. Busygin, Applying the random forest method to improve burner efficiency, *Mathematics*, vol. 10, no. 12, p. 2143, 2022.

- [208] F. Wang, S. Ma, H. Wang, Y. Li, Z. Qin, and J. Zhang, A hybrid model integrating improved flower pollination algorithm-based feature selection and improved random forest for NO_x emission estimation of coal-fired power plants, *Measurement*, vol. 125, pp. 303–312, 2018.
- [209] K. S. Madhavan, P. Prasanna, T. Varman, R. Dhanuskodi, and S. Arumugam, ANN prediction tool for ReHeater and SuperHeater sprays in boiler performance, in *Proc. 2011 3rd Int. Conf. Electronics Computer Technology*, Kanyakumari, India, 2011, pp. 335–337.
- [210] A. Adla, J.-L. Soubie, and P. Zarate, A co-operative intelligent decision support system for boilers combustion management based on a distributed architecture, *J. Decis. Syst.*, vol. 16, no. 2, pp. 241–263, 2007.
- [211] X. Zhou, J. Shen, J. X. Shen, and Y. G. Li, New immune multiobjective optimization algorithm and its application in boiler combustion optimization, *J. Southeast Univ. (Engl. Ed.*), vol. 26, no. 4, pp. 563–568, 2010.
- [212] X. Zhou, Multiobjective predictive control for boiler combustion optimization, (in Chinese), *Comput. Simul.*, vol. 30, no. 11, pp. 89–94, 2013.
- [213] E. Sun, J. Xu, M. Li, G. Liu, and B. Zhu, Connected-top-bottomcycle to cascade utilize flue gas heat for supercritical carbon dioxide coal fired power plant, *Energy Convers. Manag.*, vol. 172, pp. 138–154, 2018.
- [214] G. Ding, B. He, Y. Cao, C. Wang, L. Su, Z. Duan, J. Song, W. Tong, and X. Li, Process simulation and optimization of municipal solid waste fired power plant with oxygen/carbon dioxide combustion for near zero carbon dioxide emission, *Energy Convers. Manag.*, vol. 157, pp. 157–168, 2018.
- [215] J. Zhou, M. Zhu, K. Xu, S. Su, Y. Tang, S. Hu, Y. Wang, J. Xu, L. He, and J. Xiang, Key issues and innovative double-tangential circular boiler configurations for the 1000 MW coal-fired supercritical carbon dioxide power plant, *Energy*, vol. 199, p. 117474, 2020.
- [216] Q. Li, The view of technological innovation in coal industry under the vision of carbon neutralization, *Int. J. Coal Sci. Technol.*, vol. 8, no. 6, pp. 1197–1207, 2021.